Author's response to reviews

Title: Utilisation of malaria preventive measures during pregnancy and birth outcomes in Ibadan, Nigeria

Authors:

Olukemi O Tongo (ktongo2@yahoo.com)
Adebola E Orimadegun (beorimadegun@yahoo.com)
Olusegun O Akinyinka (asegun@hotmail.com)

Version: 6 Date: 29 July 2011

Author's response to reviews:

Response to Executive editor’s comments
1. Informed consent has been documented in the manuscript under data collection
2. Formatting of the tables have been done to conform to journal style ie cell borders have been delineated with visible black lines, Table 6 has been changed to portrait orientation.

Competing interests, author contributions and acknowledgement have been written in the order specified by the journal

Capitalization of headings have been corrected